Author's response to reviews

Title: The impact of drought on the association between food security and mental health in a nationally representative Australian sample

Authors:

Sharon Friel (sharon.friel@anu.edu.au)
Helen Berry (helen.berry@canberra.edu.au)
Huong Dinh (Huong.dinh@anu.edu.au)
Lean O'Brien (Lean.O'Brien@canberra.edu.au)
Helen L Walls (Helen.walls@anu.edu.au)

Version: 4 Date: 3 October 2014

Author's response to reviews:

Thank you to the reviewer from previously helpful comments. We understand he has no further comments. As requested by the Editor, we have noted in the section describing the data that the ANU has an Organisational License to use the HILDA data and describe the ownership of the data in a little more detail.